Author’s response to reviews

Title: Training in Dual Diagnosis Interventions (The COMO Study): A Randomised Controlled Trial

Authors:

   Elizabeth C Hughes (lhughes@lincoln.ac.uk)  
   Shamil Wanigaratne (shamil.wanigaratne@slam.nhs.uk)  
   Kevin Gournay (keyingournay@aol.com)  
   Sonia Johnson (s.johnson@ucl.ac.uk)  
   Graham Thornicroft (graham.thornicroft@iop.kcl.ac.uk)  
   Emily Finch (emily.finch@slam.nhs.uk)  
   Jane Marshall (jane.marshall@slam.nhs.uk)  
   Neil Smith (n.smith@iop.kcl.ac.uk)

Version: 5 Date: 17 January 2008

Author’s response to reviews:

I have made the necessary minor corrections as recommended by Carl Lombard, and have addressed the issue of ensuring that the baseline differences in depersonalisation in the control group did not affect the outcomes of the study